Author's response to reviews

Title: Quantitative pedigree analysis and mitochondrial DNA sequence variants in adults with cyclic vomiting syndrome.

Authors:

Thangam Venkatesan (tvenkate@mcw.edu)
Nilay Kumar (nikumar@mcw.edu)
Jyotirmoy Sengupta (rsengupta@nrconsults.com)
Muhammad Ali (muhammad.ali@phhs.org)
Baber Malik (babermmalik@gmail.com)
Aniko Szabo (aszabo@mcw.edu)
Miranda A.L. van Tilburg (tilburg@med.unc.edu)
Essam A. Zaki (ezaki@chla.usc.edu)
Richard G. Boles (rboles@chla.usc.edu)

Version: 2

Date: 12 February 2014

Author's response to reviews:

Dear Dr. Morawska,

We are submitting the revised manuscript entitled, “Quantitative pedigree analysis and mitochondrial DNA sequence variants in adults with cyclic vomiting syndrome.” for your consideration for publication.

The study was approved by the Institutional Review Board at the Medical College of Wisconsin, University of North Carolina and the Children’s hospital of Los Angeles. Informed consent was obtained from all subjects and a legal guardian when applicable. This was inadvertently placed in the statistical section and we have now included this in the methods section (page 7, lines 6-9). The revised manuscript has been uploaded.

Sincerely,

Thangam Venkatesan